Author's response to reviews

Title: Development and evaluation of a youth mental health community awareness campaign - The Compass Strategy

Authors:

Annemarie Wright (annemarie.wright@mh.org.au)
Patrick D McGorry (mcgorry@ariel.unimelb.edu.au)
Meredith G Harris (harrismg@unimelb.edu.au)
Anthony F Jorm (ajorm@unimelb.edu.au)
Kerryn Pennell (pennell@unimelb.edu.au)

Version: 4 Date: 8 August 2006

Author's response to reviews: see over
8th August 2006

Dear Editorial Team Members,

Re. “Development and evaluation of a youth mental health community awareness campaign - The Compass Strategy” ID No. 1661418290465927

Thank you for accepting, in principle, this manuscript for publication in BMC Public Health. Outlined below are the formatting changes requested in your email of 3rd August 2006.

1. Table 1 has been included in the text under a new sub-heading in the Methods section entitled “Population data sources” (see page 6). Table references and numbering of references have been amended accordingly. Please note the Ethics section has been moved to the end of the Methods section (see page 19) to account for this change in structure of the text.

2. The references section has been moved as requested.

Please let me know if you require any further information.

Yours faithfully,

Prof. Anthony Jorm

Orygen Research Centre
University of Melbourne
Locked Bag 10
Parkville
Victoria 3052
Australia